Author's response to reviews

Title: Comparison of hemodynamic and nutritional parameters between older persons practicing of physical activity, nonsmokers and ex-smokers.

Authors:

   Cristina O Francisco (cristinaft05@gmail.com)
   Natalia A Ricci (natalia_ricci@hotmail.com)
   Marcelo N Rebelatto (m_rebelatto@hotmail.com)
   José R Rebelatto (rubens@ufscar.br)

Version: 8 Date: 18 October 2010

Author's response to reviews: see over
Brazil, October 19, 2010.

To the editors of the journal BMC Geriatrics

MS:1398899350334917
Title: Comparison of hemodynamic and nutritional parameters between older persons practicing regular physical activity, nonsmokers and ex-smokers.

The manuscript was submitted for review. Thank you for your attention to our work. We place ourselves at your entire disposal for resolving any other questions or implementing any further changes.

Sincerely

Cristina Oliveira Francisco